Author's response to reviews

**Title:** Physical activity is associated with a low prevalence of musculoskeletal disorders in the Royal Norwegian Navy: a cross sectional study

**Authors:**
- Tone Morken (tone.morken@isf.uib.no)
- Nils Mageroy (nils.mageroy@isf.uib.no)
- Bente E Moen (bente.moen@isf.uib.no)

**Version:** 4 **Date:** 25 June 2007

**Author's response to reviews:**
We have revised the manuscript according to the comments of 21.06.07